Author's response to reviews

Title: Age at menarche in Canada: results from the National Longitudinal Survey of Children & Youth

Authors:

Ban Al-Sahab (bsahab@yorku.ca)
Chris I. Ardern (cardern@yorku.ca)
Mazen J. Hamadeh (hamadeh@yorku.ca)
Hala Tamim (htamim@yorku.ca)

Version: 6 Date: 19 November 2010

Author's response to reviews:

Only Figure 2 has been revised. The overall rate of early menarche in Canada was misplaced.